Author's response to reviews

Title: The association between Toll-like receptor 2 single-nucleotide polymorphisms and hepatocellular carcinoma susceptibility

Authors:

Xie Junjie Dr. (xiejunjiefabrice@126.com)
Jiang Songyao Dr. (jiangsydr@hotmail.com)
Shi Minmin Dr. (rightoleft@yahoo.com.cn)
Song Yanyan Dr. (syyssmu@126.com)
Shen Baiyong Dr. (shenbaiyong@medmail.com.cn)
Deng Xiaxing Dr. (deng_xx523@yahoo.com.cn)
Jin Jiabin Dr. (jinbin127@hotmail.com)
Zhan Xi Dr. (zhanxi80@hotmail.com)
Chen Hao Dr. (haochendr@yahoo.com.cn)

Version: 2 Date: 12 December 2011

Author's response to reviews:

Dear Editors

We submit our revised article titled as “The association between Toll-like receptor 2 single-nucleotide polymorphisms and hepatocellular carcinoma susceptibility” to BMC Cancer. We have revised our paper according to the comments of the reviewers and answered all the questions. I wish this manuscript will be fitted to the requirement of your journal. Thank you very much.

Your sincerely
Chen Hao
Department of Surgery,
Ruijin Hospital,
Shanghai Jiaotong University School of Medicine,
197 Ruijin Er Road
Shanghai, P.R. China